Author’s response to reviews

Title: Measuring implementation in Global Mental Health: Validation of a pragmatic implementation science measure in Eastern Ukraine using an experimental vignette design

Authors:

Emily Haroz (eharoz1@jhu.edu)
Paul Anthony Bolton (pbolton1@jhu.edu)
Amanda Nguyen (ajnguyen@virginia.edu)
Catherine Lee (clee147@jhu.edu)
Sergey Bogdanov (dr.bogdnaov.s@gmail.com)
Judith Bass (jbass1@jhu.edu)
Namrita Singh (namrita.singh@jhu.edu)
Samuel Doty (s.benjamin.doty@jhu.edu)
Laura Murray (lmurra15@jhu.edu)

Version: 3 Date: 08 Apr 2019

Author’s response to reviews:

April 8, 2019

Dear Dr. Reeves,

Thank you for giving us the opportunity to revise and resubmit our article "Measuring implementation in Global Mental Health: Validation of a pragmatic implementation science measure in Eastern Ukraine using an experimental vignette design" (BHSR-D-18-02532R1). We have made revisions in response to the feedback provided which are provided in track changes the manuscript. We have provided a clean version as well. Please also find our response to the reviewers itemizing these changes below.
Sincerely,

Emily E. Haroz, PhD
Department of Mental Health
Johns Hopkins Bloomberg School of Public Health

Editor Comments:

1. Consent to Participate

In your “Ethical Approval and Consent to Participate” section of your Declarations please clearly state whether the consent obtained from the participants was written or verbal. If verbal, please state the reason and whether the ethics committee approved this procedure.

RESPONSE: We apologize and had only added this in the text. Consent for both the qualitative and quantitative studies was verbal so as to ensure confidentiality. This process was approved by both IRBs. This was added to the Declarations section as well.

3. Figure 1

Thank you for providing the source of the map in Figure 1. Please include the source in your reference list and then cite it in the Figure Legend.

RESPONSE: We have added the citation in the reference list and in the figure legend

6. Clean Manuscript
Thank you for uploading a clean version of your manuscript, please again upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files. Please also ensure that all figures, tables, and supplementary/additional files are cited in the text.

RESPONSE: We have uploaded a clean version